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Still some 2dvd problems

Submitted by Patrick.Gatlin on Sat, 2013-05-04 09:20
Date:
Saturday, May 4, 2013 - 08145

Instrument List:

* Yellow
Discussion:
SN70 stopped collecting data at 03:07 UTC (data collection restarted at 1330 UTC)
SN38 stilldown due to video camera problems

SN37 running properly with no data outages (4 May rainfall total: 22.7 mm of rain through 1300
UTC)

SN36 - no remote comms

SN35 - running properly with no data outages (4 May rainfall total: 16.2 mm of rain through
13:57 UTC)

SN25 stopped collecting data late Friday (2 May), but looks like cameras collected data so it
may be a synchronization issue. | will ask Joanneum we can patch together the missing HYD
file(s) from the past 2 days.

A site visit will be done to SN36 and SN25 today to further investigate the problems and get
data collecting going again.



